Author’s response to reviews

Title: Hepatic tuberculosis presenting with extremely high serum ferritin misdiagnosed as adult onset Still's disease - A Case report

Authors:

E.M. Manoj EMM (manojediri@yahoo.com)
R Srigrisna RS (rsrigrishna@yahoo.com)
Murugapillai K Ragunadan MKR (muruguragu@gmail.com)

Version: 2 Date: 28 December 2011

Author’s response to reviews: see over
Author's response to the review.

Title: Hepatic tuberculosis presenting with extremely high serum ferritin misdiagnosed as adult onset Still's disease - A Case report

Authors:

E.M. Manoj (manojediri@yahoo.com)

R.Srigrishna (rsrigrishna@yahoo.com)

Murugapillae K. Ragunathan (muruguraqu@gmial.com)

Version: 2 Date: 28 December 2011

Author's response to reviews: see over
28.12, 2011.

The Biomed Central Editorial Team

Object: MS ID : 2146290045654550, Hepatic tuberculosis presenting with extremely high serum ferritin misdiagnosed as adult onset Still’s disease - A Case report. E.M Manoj et al

Thank you for consideration of our manuscript for publication in your journal. We have reviewed the above manuscript according to your comments and apologize for mistakes made.

01. Please indicate and explain in your cover letter how the case report adds to the medical literature using one or more of the criteria below:

Unreported or unusual side effects or adverse interactions involving medications.

Unexpected or unusual presentations of a disease.

New associations or variations in disease processes.

Presentations, diagnoses and/or management of new and emerging diseases.

An unexpected association between diseases or symptoms.

An unexpected event in the course of observing or treating a patient.

Findings that shed new light on the possible pathogenesis of a disease or an adverse effect.
The reported case of **Hepatic tuberculosis presenting with extremely high serum ferritin misdiagnosed as adult onset Still’s disease - A Case report** is an unexpected or unusual presentations of a disease.

02. A spelling mistake in the topic written as “adilt” was corrected as “adult”.